Author's response to reviews

Title: How Adolescents Perceive Their Communities: A Qualitative Study That Explores the Relationship Between Health and the Physical Environment

Authors:

Kristin N Mmari (kmmari@jhsph.edu)
Hannah Lantos (hannah.lantos@gmail.com)
Heena Brahmbhatt (hbrahmbh@jhsph.edu)
Sinead Delany-Moretlwe (sdelany@wrhi.ac.za)
Chaohua Lou (chaohual@yahoo.com)
Rajib Acharya (racharya@popcouncil.org)
Adesola Sangowawa (daisyolu@yahoo.co.uk)

Version: 2
Date: 19 December 2013

Author's response to reviews:

Thank you very much for ensuring our manuscript has all the necessary components. As requested, we have removed both our title page and authors contributions and have now put them in the main manuscript. We have also added a section about the authors' competing interests, and have removed our acknowledgements from the title page and placed it after the authors' competing interests, as part of the main manuscript.

Let us know if there is anything else.

Thank you, and we look forward to our manuscript getting peer-reviewed.

Sincerely,

Kristin Mmari